Page1 CS248 Lecture 17Copyright � Pat HanrahanComputer Graphics ModelsFor example, SGI shading modelTerms:nAmbient - Hack!nDiffusen

Lr = ka La + kd (N•Li )+ ks(N•Hi )s ( ) ∑ Li

CS248 Lecture 17Copyright � Pat HanrahanLighting CalculationsIllumination and reflection calculationsnNormals n Coordinate systems

Lighting calculations

are derived from

laws of physics

must be done in cs

metric to world

TransformShade

Transform

Project PPN

Page 2

CS248 Lecture 17Copyright � Pat HanrahanTransformation of NormalsNormals remain perpendicular to tangentsRotations special case!

N•V = 0 ′N • ′V = 0

CS248 Lecture 17Copyright � Pat HanrahanCubical Environment Map

Page 3

CS248 Lecture 17Copyright � Pat HanrahanGazing BallNote:nPhotograph of reflective ball n Reflection indexed by normal n Maps entire field of view to circle n

Resolution function of orientation; maximum head-on n Fish eye camera lens similar

Pat HanrahanDisplacement/Bump MappingOffset surface positionnDisplacementnPerturb normal